

JNCC Report 747

Policy interventions to encourage sustainable consumption Guidance Report

Maddie Harris

November 2023

© JNCC, Peterborough 2023

ISSN 0963 8091

JNCC's report series serves as a record of the work undertaken or commissioned by JNCC. The series also helps us to share, and promote the use of, our work and to develop future collaborations.

For further information on JNCC's report series please contact:

Joint Nature Conservation Committee Monkstone House City Road Peterborough PE1 1JY <u>https://jncc.gov.uk/</u> Communications@jncc.gov.uk

This report should be cited as:

Harris, M.¹ 2023. Policy interventions to encourage sustainable consumption. *JNCC Report* 747 (*Guidance report*), JNCC, Peterborough, ISSN 0963-8091. https://hub.jncc.gov.uk/assets/859c5573-1f84-4122-b509-65d779489757

Author affiliation:

¹ JNCC, Monkstone House, City Road, Peterborough, PE1 1JY

Acknowledgments:

We would like to thank Scottish Government, Defra, Welsh Government and DAERA for conception of the idea for this report, steering group representation and review. We would also like to thank WWF for their review and input.

Evidence Quality Assurance:

This report is compliant with JNCC's Evidence Quality Assurance Policy https://jncc.gov.uk/about-jncc/corporate-information/evidence-quality-assurance/

The views and recommendations presented in this report do not necessarily reflect the views and policies of JNCC [or of the other collaborating Departments and agencies].

This report and any accompanying material is published by JNCC under the <u>Open</u> <u>Government Licence</u> (OGLv3.0 for public sector information), unless otherwise stated.

Summary

Sustainable consumption is increasing in profile as an area of policy priority. This report synthesises key policy interventions that could be used by governments to influence the sustainability of consumption (Figure 1). These must target a range of different supply chain actors, including individual consumers, producers, processors, importers, retailers, investors and governments. In order to improve the sustainability of consumption, each of these actors needs access to sustainable options, an understanding of how sustainable each option is, encouragement to choose the most sustainable options, prevention from choosing the most harmful options and, where appropriate, support to reduce total levels of consumption. Policy interventions that can be used to enact this change can be grouped into infrastructure-based interventions (which include sustainable waste disposal and a circular economy, and supporting local consumption, where appropriate), information-based interventions (including ecolabelling, certification, awareness raising campaigns, school education, and capacity building), economic interventions (such as taxes, subsidies, and funding for research and implementation of sustainable options) and regulatory interventions (such as through free trade agreements, multi-lateral agreements, bans or quotas, sustainable public procurement rules, and controls on advertising). There is significant crossover and synergy between each of these policy interventions, with no one intervention able to create as significant a change alone as when undertaken alongside others. A suite of different policy interventions targeting actors across the supply chain will therefore be most effective at increasing the sustainability of consumption.



Figure 1. A summary of policy interventions that could be used by governments to improve the sustainability of consumption, the actors they can target and the barriers they can help to break down.

Contents

1	Introduction		1	
	1.1	Policy context	1	
	1.2	Aims and scope	1	
	1.3	Disclaimer	2	
2	Wh	ich actors can sustainable consumption policy interventions target?	3	
	2.1	Individual consumers	3	
	2.2	Producers and processors	3	
	2.3	Businesses, traders and retailers	4	
	2.4	Investors and financial institutions	4	
	2.5	Governments	4	
3 What do these actors need to improve the sustainability of consumption?			5	
	3.1	Access to sustainable options	5	
	3.2	An understanding of the sustainability of options	5	
	3.3	Encouragement to choose sustainable options	5	
	3.4	Removal of the least sustainable options	5	
	3.5	A reduction in overconsumption	6	
4				
su	stain	ability of consumption?	7	
	4.1	Infrastructure-based interventions	7	
	4.2	Information-based interventions	9	
	4.3	Economic interventions1	3	
	4.4	Regulatory interventions 1	5	
5	Cor	nclusions	0	
Re	References			
Ar	Annex 1: Methods			

1 Introduction

This report aims to summarise policy interventions that could be used by governments to influence the sustainability of consumption. It begins by introducing readers to the policy context and scope for the document (Section 1). It then explores audiences that could be targeted by sustainable consumption policy interventions (Section 2). Next, it identifies key barriers to sustainable consumption that policy interventions could aim to overcome (Section 3). Finally, it runs through a range of possible policy interventions grouped by type, giving examples of where these have been implemented and, where information is available, commenting on the effectiveness of their implementation and other factors that would need to be considered (Section 4).

1.1 Policy context

Sustainable consumption is important, both to prevent environmental impacts such as biodiversity loss, and also to ensure continued resource security and supply chain resilience. It is increasing in profile as an area of policy priority; the issue has been highlighted recently in multiple high-profile reports, such as the <u>National Food Strategy</u>, the <u>Dasgupta review</u> and the <u>first draft of the Convention on Biological Diversity's post-2020 framework</u>. Across the four UK countries, the problem is being recognised in key policy documents:

- England's <u>25 Year Environment Plan</u> aims to "avoid improving our domestic environment at the expense of the environment globally"
- Wales' 2015 <u>Wellbeing of Future Generations Act</u> has a goal for "a globally responsible Wales," which includes "ensuring that our supply chains are fair, ethical and sustainable," "supporting sustainable behaviour," and "efficient use of resources"
- Scotland's 2020 <u>Environment Strategy</u> includes an outcome that "we are responsible global citizens with a sustainable international footprint"
- Northern Ireland's <u>Sustainable Development Strategy</u> has a guiding principle of "living within environmental limits"

<u>Recent JNCC work</u> (Croft *et al.* 2021) has progressed our understanding of the environmental impacts that the UK has through its consumption, through the development of a <u>national indicator</u>. This can be used as a tool to prioritise the sectors and geographies where interventions are most likely to make a difference. However, the workstream has not yet explored what interventions could be carried out. This report therefore aims to identify policy interventions that could be used to improve the sustainability of a country's consumption.

1.2 Aims and scope

The report is based on a review of both scientific and grey literature to identify policies that have been undertaken in other countries, as well as policies that have been suggested as solutions in the scientific literature even if not yet implemented. The review was time-restricted to five days and so did not aim to be comprehensive or systematic.

The scope of the review was limited to actions that could be undertaken by governments. Therefore, actions that could be undertaken directly by (for example) individual consumers were excluded; however, actions that the government could take to engage with, educate or legislate in ways that might change the behaviour of those consumers were included. The report does not aim to give the full detail that would be required for implementation of these policy interventions, but rather aims to frame thinking with regards to the options that are available and that could be considered.

Whilst previous JNCC work in this area has largely focused on agricultural supply chains, this report has a broader sectoral scope, including policy interventions of relevance to any sector of the economy. The review does not constitute a 'deep dive' into any one environmental impact type (e.g. biodiversity, GHG emissions) or commodity to identify policy interventions specifically applicable in that case. Instead, it focuses on generic and cross-cutting policy interventions that could be applied to reduce any environmental impact type of interest. However, it is important to highlight the importance of and need for additional complementary research that focuses in more detail on specifics.

1.3 Disclaimer

This report does not aim to make specific recommendations about which policy interventions should be adopted; nor does it constitute any kind of suggestion or commitment that any of the policy interventions mentioned will be implemented by a particular actor. Rather, it aims to act as an information source that could be used, alongside other information, by policy makers who are taking such decisions. JNCC is an independent body with a remit for providing impartial expert advice on nature and conservation, with no involvement in policy implementation.

2 Which actors can sustainable consumption policy interventions target?

2.1 Individual consumers

In general, goods produced within the economy are ultimately 'consumed' by individuals, whether directly (e.g. buying an apple to eat, or buying fuel to use in your car) or in an embedded form (e.g. soy used to feed the cow that was made into a burger that you buy, or the fuel in the factory that was used to make your car). Policies that support individuals to consume more sustainably through influencing behaviours such as purchasing decisions and household waste production therefore have the potential to be wide-reaching. For example, where individuals lack knowledge on the impact of their purchasing, they will be unable to take this into account when making decisions. Similarly, where unsustainable options are more affordable than sustainable ones, this price difference may influence purchasing decisions (Jonkute-Vilke & Staniškis 2019; Pape et al. 2011). In these cases, governments could help through information provision (see Section 4.2) or use of economic policy interventions (see Section 4.3). Infrastructure based interventions (see Section 4.1) and regulatory interventions (see Section 4.4) that help ensure sustainable options are the 'default' could also be helpful. However, a consumer's access to sustainable options is dependent on other actors in the supply chain (e.g. producers, businesses), and so policies aimed at individuals alone will likely be ineffective.

2.2 **Producers and processors**

The locations at which the environmental impacts associated with consumption take place are often far removed from the location at which the goods are consumed. For example, the deforestation associated with soy occurs where the soy is grown (Pendrill et al. 2019); the pollution associated with metals happens where these are extracted (Gosar 2004); and much of the water stress associated with textile production comes about where materials are processed and dyed (Raja et al. 2019). Policy interventions influencing producers and processors (such as providing funding for research on and implementation of sustainable practices, see Section 4.3.3) may therefore influence sustainability of the system overall in a more direct manner (Stevens 2010). This may make it easier to identify, to understand the impacts of and to target solutions, thereby solving specific problems on-the-ground. Although the purpose of this review is to identify policy interventions to improve the sustainability of consumption rather than the sustainability of production, any changes in production impacts at a large scale will feed along the supply chain to influence consumption impacts also (if we assume that other factors, such as total volume of consumption, remain the same). Therefore, any framework of policies aiming to influence the sustainability of consumption will need also to consider the sustainability of production as an integral component. Policies aimed at consumption ultimately aim to influence production and processing practices through changing demand.

Targeting producers in addition to consumers also reduces the risk of displacement, which is a concern when policies are aimed at consumers alone. This is the problem that if consumers attempt to improve their sustainability by simply shifting supply from an area with unsustainable production to one with sustainable production, that another consumer will simply take on the unsustainable supply that is left behind, thereby leading to no actual change in the system. However, if producers are targeted as well, unsustainable supply will be addressed whether there is a different potential customer or not.

2.3 Businesses, traders and retailers

Most consumers do not purchase goods directly from producers or processors, but from businesses and retailers that act as a 'middleman'. Traders may in turn act as 'middlemen' between producers and retailers. These actors therefore play an important role in terms of what is made available to consumers. They may have a more direct (and coordinated) influence on producers and processors than individual consumers do (although supply chains can be long and complex, so whilst more direct than individual consumers they are often still far removed). Businesses are increasingly looking for opportunities to show their green credentials to consumers and investors as a selling point (Neill 2020). Policy interventions that help to encourage this further, such as supporting ecolabelling and certification (see Section 4.2.1), can therefore play a role in improving the sustainability of consumption. It is important for any policy targeting businesses to ensure that greenwashing (the use of green credentials that are not backed up by evidence) is prevented.

2.4 Investors and financial institutions

Investors and financial institutions are integral to many stages of the supply chain. If there is no investment in a particular product, it will be much harder for this product to reach markets and be consumed by individuals. As the key aim of investors and financial institutions is to make money from their investments, any economic policy interventions (see Section 4.3) that support the market competitiveness of sustainable options over unsustainable options may be particularly effective. There is also a growing interest in ethical investments (Martini 2021), suggesting that information-based interventions (see Section 4.2) that are aimed at investors and financial institutions to help them understand the impacts associated with their investments, could also be of increasing use. This highlights the importance of ensuring that information-based interventions are targeted to multiple audiences, with adapted material to be of most relevance to each case. For example, an investor or financial institution may not automatically make the link between an awareness campaign aimed at individual consumers about how this could be useful in other contexts.

2.5 Governments

Governments can also implement policies aimed at themselves, or at the governments of countries with which they work. For example, public procurement (see Section 4.4.4) constitutes a significant proportion of a country's total consumption. For example, in OECD countries, it makes up 12% of GDP on average (UNEP 2017). Ensuring this is spent responsibly can be used to both show leadership as an awareness raising tool, and to help drive markets towards sustainable options (BIO Intelligence Service 2012; UNEP 2015). Governments can also work with other governments through interventions such as trade agreements (see Section 4.4.1), which can have an effect on which goods are able most competitively to make it to market, and multi-lateral environmental agreements (see Section 4.4.2), which can provide a united message to encourage changes in production practices more widely. There are also other international governance systems in place that should be considered, such as the World Trade Organisation.

3 What do these actors need to improve the sustainability of consumption?

3.1 Access to sustainable options

If actors are not able to access sustainable options, they are unable to consume, produce or invest sustainably. For example, if producers or processors do not have investment for research or implementation to improve the sustainability of their practices (see Section 4.3.3), they may have no option but to continue their current practices. In turn, this would mean that businesses and retailers cannot pass on sustainable produce to final consumers. If consumers are being encouraged to adopt solutions such as consuming locally, but do not have access to local produce through the infrastructure available (see Section 4.1.2), they will find it difficult to follow the advice even if they wished to. If unsustainable products are more competitive in the market than sustainable alternatives, it is less likely that investors, businesses and retailers will take the risk to supply them as an option to final consumers (see Section 4.3 on economic interventions). Interventions aimed at breaking down these barriers to access could be a useful first step to ensure that sustainable options are available. Although alone this may do little, it is necessary before interventions aimed at increasing the likelihood that these options will be selected by consumers can be implemented. Combining actions aiming to improve access to sustainable options with actions aiming to improve the likelihood of selection of sustainable options could have a much greater impact than improving access alone.

3.2 An understanding of the sustainability of options

If sustainable options are available, but actors do not understand the relative impacts of the different options, there is no reason for them to select the sustainable options over other options. If information on the impacts linked to consumption is provided (see Section 4.2), actors who wish to are able to make informed decisions about their production practices, investment decisions or purchasing choices. Whilst this may represent a small proportion of total actors, public interest in the area is growing (Deloitte 2022). Whilst other barriers such as price and ease of access may have a greater impact on an actor's decisions, if used in combination with interventions aiming to break down those barriers then information provision can be an important step (Deselnicu *et al.* 2014; Pape *et al.* 2011; Scholl *et al.* 2010).

3.3 Encouragement to choose sustainable options

Governments can also use a range of interventions aimed at encouraging actors to select sustainable options by improving the ease of access to them and reducing the ease of access to unsustainable options, through influencing 'default' behaviours. For example, this could be through provision of infrastructure (see Section 4.1) or through intervening in affordability through taxes or subsidies (see Section 4.3). Whilst interventions aimed at improving understanding will typically only affect actors who are already interested in the problem, interventions aimed at encouraging sustainable choices may be effective in steering even disinterested actors towards more sustainable choices. They can therefore be more comprehensive in who they affect.

3.4 Removal of the least sustainable options

In cases where there is strong agreement that an option has significantly negative environmental consequences, there may be scope to ban it, or to set a quota (see Section 4.4.3 – and note references in that section to WTO rules). This would remove the option from the market, so consumers are unable to select it. Assuming their other consumption remains consistent, and they are forced to find a more sustainable alternative to the option in question, this would improve the sustainability of their consumption overall. Bans could also be made in the context of a smaller scope, for example through setting rules around public procurement (see Section 4.4.4) rather than at a national scale, or through making bilateral agreements with individual trading partners (see Section 4.4.1) to restrict what is imported from a single producing country rather than across the board.

3.5 A reduction in overconsumption

The environmental sustainability and resource security associated with consumption is not only linked to the sustainability of products that are consumed, but also to how much is consumed overall. Even if the production of a particular good in isolation has minimal effects on the environment or on the system's ability to regenerate and provide resource security, production of this good at a large scale can multiply those minimal effects to significant effects. The total amount of consumption in a country is related to individuals' purchasing decisions, wealth and population levels. There is a strong imbalance in levels of consumption around the world, meaning that whilst interventions relating to overconsumption may be apt to consider in countries with high levels of consumption, other countries will need to focus on ensuring they can support an increased level of consumption to support growing populations and reductions in poverty. Preventing overconsumption is often seen as the least popular policy intervention due to the importance placed on GDP and economic growth (Lorek & Fuchs 2013; Pape et al. 2011). However, some authors highlight the importance of more holistic economic metrics that do not only measure current economic activity, but also reflect how it might impact future potential for growth (UNEP 2015), for example through inclusion of natural capital accounts. Similarly, if consumption practices are undermining resource security, this prevents potential for future growth. Interventions that could help to prevent overconsumption include supporting a service-based economy that may allow for growth with minimal environmental impact (see Section 4.1.2), ensuring information-based interventions (see Section 2) address the risks of overconsumption alongside the risks of other types of unsustainable consumption, implementing quotas on the least sustainable products (see Section 4.4.3), and supporting access to family planning services.

4 Which policy interventions can be used by governments to improve the sustainability of consumption?

4.1 Infrastructure-based interventions

Infrastructure can be very influential in terms of consumer behaviour. 'Green Growth' strategies typically focus on investment in sustainable infrastructure. These may include options related to bike lanes or public transportation (not explored here) to improve the sustainability of transport options, or infrastructure related to ensuring the provision of sustainable waste disposal (see Section 4.1.1) and purchasing options (see Section 4.1.2).

4.1.1 Sustainable waste disposal and a circular economy

Anything that is produced has environmental impacts, for example through putting pressure on land, water and biodiversity. Reducing waste could allow for consumption needs to be met with a lower volume of production, and therefore lower impacts. For example, it is estimated that 30–40% of all food produced for human consumption is lost or wasted (FAO 2011; UNEP 2021; WWF UK 2021). If this could be minimised, it could represent a significant reduction in the total environmental impacts associated with food production, as well as addressing social issues such as food security (UNEP 2021). Unsustainable waste disposal also causes problems in terms of pollution. For example, it is estimated that about a third of all plastic packaging produced ends up dispersed in the environment rather than collected (WRAP 2022). Improving waste disposal infrastructure and encouraging a circular economy can help reduce both the total amount of waste, and the total amount that is disposed of unsustainably. The <u>Ellen Macarthur Foundation</u> has done a lot of work on circular economy principles, including sharing products, maintaining and prolonging the life of products, reusing products, refurbishing products, recycling products and returning the nutrients from biodegradable materials to the earth.

In the UK, waste disposal is a devolved policy area.

4.1.1.1 Kerbside recycling collection and waste separation

Provision of infrastructure to support recycling gives people access to an option to dispose of their waste more sustainably, and therefore reduce the associated environmental impacts. Kerbside recycling in particular encourages households to choose to recycle by making it a simple and convenient option that does not require the time, expense and behavioural change that drop-off points would, especially if combined with other interventions such as education and economic incentives (Struk 2017).

Kerbside recycling measures are already widely practised (Scholl *et al.* 2010), but significant scope for improvement remains. For example, it is estimated that only 9% of plastics ever produced have been recycled (Geyer *et al.* 2017). One example identified within the literature of where such measures have been implemented is the Catalonian organic waste regulations (Wolff *et al.* 2017). These aimed to reduce disposal in landfills through separate collection of organic waste. Sixteen years after its introduction, 63% of municipalities had implemented it, falling short of targets (Wolff *et al.* 2017). When combined with a tax on those that had failed to implement it after this point, compliance increased considerably. No direct assessment was made on the environmental impacts of this initiative (Wolff *et al.* 2017).

As well as improving infrastructure at the point of collection, this intervention relies on the provision of infrastructure able to process the recyclable waste. Currently, much of the UK's waste is sent abroad for recycling, leading to concerns around associated environmental and social implications, as well as longevity of the solution given some countries are starting to reject further waste imports (Burgess *et al.* 2021). Another concern raised in relation to interventions, such as recycling, that aim to improve resource efficiency is the 'rebound effect.' This is where consumers worry less about the costs or impacts of the goods they consume and the waste they produce, because they feel that solutions are already in place to mitigate the impacts they are having (UNEP 2012).

4.1.1.2 Deposit refund schemes

Deposit refund schemes combine infrastructure provision with an economic incentive for consumers to return recyclable or reusable items, such as glass bottles, batteries, tyres, oil and electronics. The incentive does not involve funding provision by the state, but rather is based on a system where the consumer pays an extra fee when they purchase an item, that can be refunded when it is returned. Studies have shown that such schemes can be very effective at increasing return rates, but that associated administrative costs can by fairly high (UNEP 2015). Although used in many countries, one example found in the literature review is the 'Packaging Ordinance with deposit scheme' that was implemented in Germany from 2006. This required consumers to pay at least 0.25 per unit when buying single-use plastic bottles and other drink packaging. The scheme was successful in significant improvements in waste disposal of such items, but did not achieve its principal aim of stabilising the share of more sustainable drinks containers on the market (Wolff *et al.* 2017). As with the case of kerbside recycling, this intervention is aimed at the consumer, and will only be effective if infrastructure is subsequently available effectively to recycle or reuse the collected waste.

4.1.1.3 Encouraging a service-based economy

One way to reduce the impacts of consumption in a way that does not threaten economic growth could be a shift in perspective from consumption being largely about purchasing new products, to having a greater focus on purchasing services. This could take the form of aftersales services, whereby companies selling a product are mandated to provide long-term warranties or repair services, or rental-based business models, whereby consumers could pay to use an item they will only need rarely, rather than buy one themselves (UNEP 2015). The sale of second-hand goods could also be encouraged.

Governments could support the shift towards a more service-based economy through provision of infrastructure (e.g. requiring all shopping centres to ensure that a certain percentage of their area is used for repairs), through information campaigns communicating its advantages, through economic support (e.g. subsidising rental-based business models) or regulatory factors such as mandating long-term warranties (UNEP 2015).

One example of a government initiative aiming to encourage the shift towards a more service-based economy is the "Sharing City, Seoul" project, launched in 2012 (Urban Sustainability Exchange 2014). This includes a wide range of programmes and policies aiming to solve a range of the city's economic, social and environmental issues through the construction of 'secondary' infrastructure and creation of new economic opportunities (UNEP 2015). The project defined primary infrastructure as that already covered by existing policies, such as roads, schools, and libraries. Meanwhile, secondary infrastructure was considered to be that for which opportunities existed but for which a lack of infrastructure prevented communal use, to increase the sharing of objects, spaces and talents (Urban Sustainability Exchange 2014). The project has been reported on as a success, with other cities trying to

follow Seoul's example and set up similar projects (Urban Sustainability Exchange 2014), but no data were found directly assessing the environmental impacts of the scheme.

Service-based approaches will not be relevant to all sectors. For example, food and other one-use items will not be able to follow such a model. However, for many other sectors, such as clothing, transport and home goods, it could create significant opportunities.

4.1.2 Supporting local consumption, where appropriate

For certain products, consuming goods produced locally can be beneficial in terms of reduced greenhouse gas emissions from transport and a greater certainty in the environmental (and social) impacts associated with their production. Therefore, support for the provision of infrastructure such as local markets, and provision of space for them in popular locations, is suggested in the literature as a useful policy intervention to provide access to local produce (UNEP 2015). It is important to note that not everything is more efficient to produce locally (life cycle assessments can report on the differing impacts between production systems and should be used if aiming to make direct comparisons between two different options) and so further research should be undertaken if aiming to implement such a policy in a more targeted way or to assess its success. For example, one study found that whilst potatoes, beef and apples led to lower carbon emissions in the UK compared to when imported, that this was not the case for tomatoes, strawberry, poultry and lamb (even when considering carbon emitted during transport), when comparing to a selection of alternative producer countries (Webb *et al.* 2013).

In the UK, this could be implemented at any scale (national, devolved, or local government).

4.2 Information-based interventions

Information-based interventions focus on communicating the impacts of consumption to allow actors to make more informed choices. These were found to be the most commonly referenced type of intervention that came up as part of the literature review, consistent with findings from a previous study (Wolff et al. 2017). With a lack of clarity in the current information available to supply chain actors, knowledge can present a significant barrier to making responsible choices for those who wish to do so. However, studies investigating the effectiveness of information-based interventions had mixed results (Deselnicu et al. 2014), with many suggesting limited willingness for consumers to change their consumption patterns based on greater knowledge of environmental impacts (e.g. Upham et al. 2011) and others showing greater potential (e.g. Testa et al. 2015). Many concluded that they were only effective when combined with other types of intervention that focus on factors such as price or health, which can be more influential to consumer behaviour than knowledge of environmental impacts alone, a phenomenon known as the 'value-action gap' (Deselnicu et al. 2014; Pape et al. 2011; Scholl et al. 2010). In addition, information-based interventions could help change norms and provide greater acceptance for other policy interventions such as taxes and bans (see sections 4.3 and 4.3) which may otherwise prove to be unpopular due to their more invasive nature (Röös et al. 2021).

4.2.1 Ecolabelling and certification

Ecolabelling consists of adding labels to products, so consumers have more information available at the point of purchase about the impacts that a product they are purchasing has had. The theory is that this could change consumer behaviour through breaking down barriers related to knowledge. For example, an ecolabel may show the amount of carbon that has been released in the production of a particular product or the amount of water that has been used, typically through Life Cycle Assessment (a method that analyses all inputs and outputs of a defined life cycle for a particular product). The ISO (International Organization for Standardization) has produced a set of standards for ecolabels and environmental reporting more generally through the 14000 suite of standards, which includes Type I labels (awarded by national authorities based on a set of criteria), Type II labels (environmental claims made by companies themselves, for example in their advertising) and Type III labels (awarded by a qualified independent organisation such as a certification scheme body). Another example of product labelling is recycling labels, which help consumers to identify how to sustainably dispose of the waste (see Section 4.1.1) associated with products that they purchase (Mirela *et al.* 2014). Labels can either be positive (certifying that a product has been produced sustainably) or negative (signalling that a product has a large environmental footprint). Consumers are likely to be more susceptible to negative labelling but companies are more likely to be open to implementing positive labelling (Röös *et al.* 2021).

Currently, ecolabelling and certification are largely run and implemented voluntarily by companies or independent certification bodies. As well as centralising certification schemes to act as a national provider, governments could help encourage ecolabelling and certification through mandating and standardising the information that companies report to ensure consistency across labels and certification schemes. It has been shown that greenwashing (unsubstantiated or irrelevant environmental claims, made in cases where companies aim only to attract environmentally conscious consumers rather than also to improve the sustainability of their products) is a significant problem, especially in terms of consumer trust in claims, which is necessary for behavioural change to occur (UNEP 2015). One study found only 52% of sustainability claims assessed to be acceptable (Pape *et al.* 2011). Therefore, it has been suggested that another useful role Governments could play in ensuring the accurate communication of environmental product claims is through setting up a watchdog, impartial product testing, or an ombudsman function that allows consumers to complain (UNEP, 2015).

Another limitation of ecolabelling is that whilst it may help consumers make the best choice within a particular product category (e.g., between different types of fish), it does not help encourage consumers to change their consumption patterns more broadly in terms of shifting from the most harmful product categories to unrelated alternatives (e.g., from red meat to plant based proteins), which would have a greater overall impact than smaller within-category shifts (Poore & Nemecek 2018; Röös *et al.* 2021). Again, using it in combination with other interventions (such as more general awareness raising campaigns – see section 4.2.2) would therefore be important.

Similarly, ensuring that consumers understand the full picture of their consumption in a holistic manner and do not assume that buying more or bigger products labelled as sustainable will lead to no effect on the environment or offset impacts from other goods they purchase, is an important risk to mitigate (UNEP 2015). Even if all products consumed are within a threshold of sustainability, consuming more of them would lead to increased impacts overall (see Section 3.5).

It is also important for labelling schemes to consider public interpretation of information presented. For example, one study showed that the public found it very difficult to understand carbon emission values presented as part of a UK carbon labelling trial, without being provided with further information (Upham *et al.* 2011). If consumers are unable to understand the information presented, it will not lead to behavioural change.

One example of an ecolabelling scheme is the EU Ecolabel scheme. This is a voluntary label, originally set up in 1992 and subsequently revised in 2009-10, that aimed to promote resource efficiency and reduce the environmental impacts of products (Deselnicu *et al.*

2014). Products that have been awarded an EU Ecolabel show a green flower logo. Although considered key to the EU's Sustainable Consumption and Production Action Plan (Deselnicu *et al.* 2014), uptake has been relatively low, with the extent of documentation required, costs of implementation, a lack of recognition, and a lack of support from public bodies (e.g. through inclusion in sustainable public procurement criteria) cited as key barriers (Iraldo & Barberio, 2017; Prieto-Sandoval *et al.* 2020). No studies were found within the time limits of the review that went beyond exploring uptake rates to identifying specific environmental effects associated with the scheme's implementation.

In the UK, certification and ecolabelling could be implemented by either national or devolved governments.

4.2.2 Awareness raising campaigns

Awareness raising campaigns focus on a broader suite of communication tools than product labels alone. They may include consumer guides, dietary advice and declaration of footprints. This can allow for presentation of a more holistic message (e.g. demonstrating where sustainable alternatives would involve switching product category or highlighting that it is important to reduce overconsumption overall) than ecolabels alone, but information provided may be less specific. Like with ecolabels, it is important for any awareness raising campaign to consider public interpretation of information presented and how best to simplify advice without losing nuance. As with all interventions presented in the information-based interventions section, increasing knowledge needs to be carried out in combination with other policy interventions that provide easier access to sustainable options.

One risk of awareness raising campaigns, particularly when related to consumption, is that they can get 'drowned out' by companies' professional marketing campaigns promoting the increasing sale of products (UNEP 2015). This suggests potential for synergy with interventions based on advertising controls (see Section 4.4.5). It has been suggested that governments could learn much from marketing experts in terms of how to most effectively transmit information that can influence purchasing decisions (BIO Intelligence Service 2012).

Examples of public information campaigns on the theme of sustainable consumption include the Swedish Food Agency's 2017–2019 campaign to reduce food waste (Röös *et al.* 2021), Ireland's Power of One campaign to build people's understanding of the environmental impacts associated with energy consumption (Pape *et al.* 2011) and Denmark's One Tonne Less campaign which aimed to reduce CO₂ emissions by providing practical and motivational information (Scholl *et al.* 2010). Again, the impact of these campaigns remains largely unclear, with one paper commenting on the fact that evaluation reports for several such campaigns exist but are not available in the public domain (Pape *et al.* 2011). However, analysis of the One Tonne Less campaign and 17% thought it had helped them in reducing their emissions (Scholl *et al.* 2010). Factors believed to have led to this campaign's success include its careful and evidence-based planning, strong engagement from political leaders and celebrities which enhanced visibility and credibility, the building of an online community, organisation of events and workshops, and visibility at major music and sports events (Scholl *et al.* 2010).

In the UK, awareness raising campaigns could be implemented by either national or devolved governments.

4.2.3 Education in schools

Ensuring that sustainable consumption habits are taught as a cross-cutting part of the national educational curriculum across all ages is another intervention governments could use to increase consumers' knowledge in a way that could help break down barriers to behavioural change (Jonkute-Vilke & Staniškis 2019). For example, integrating sustainability into home economics and food technology classes could help to increase students' knowledge of how to put together meals that are not only sustainable, but also economic and tasty (Röös et al. 2021; Tucker 2018). For example, a lack of knowledge in how to cook appetising plant-based meals is a well-documented barrier to those who wish to reduce their meat consumption (which would make a significant difference to the land required to support consumption due to the reduced need for pasture land) from making the change (Graça et al. 2019). Similarly, technology lessons could include teaching on eco-innovation, biology could discuss endangered species, history could outline changes to the environment over time and economics could cover natural capital thinking (UNEP 2015). Governments could also support schools to run extra-curricular volunteering activities relating to sustainable consumption and ensure these are recognised through the award of credits or certifications (UNEP 2015). Activities involving 'greening' the school, such as recycling and rainwater harvesting, could be undertaken to give students further experience and inspiration (UNEP 2015). Ensuring sustainable consumption is covered in teacher training courses would help equip teachers with the relevant skills to enable this intervention to be as effective as possible (UNEP 2015).

One example of the use of education in schools as a policy intervention to promote sustainability is Ireland's Green Schools Programme (Pape *et al.* 2011). This aimed to ensure that positive environmental behaviours were integrated into both students' school lives and home lives. Analysis showed that students who had been through the programme had higher scores across all sustainability-related behaviours assessed, although longer term research would be required to demonstrate that such behavioural changes endured into their adult life (Pape *et al.* 2011).

In the UK, education is a devolved power.

4.2.4 Capacity building

Learning does not stop when people leave school. Training courses to build capacity and improve understanding of how best to make sustainable choices can be designed and targeted at any of the actors listed in Section 2 (individual consumers, producers, businesses, investors, and policy makers). Chef training to increase skills in sustainable gastronomy is a commonly cited example (e.g. Röös *et al.* 2021, Tucker *et al.* 2018). This includes increasing knowledge of how to use effectively local, seasonal, and vegetarian ingredients.

One example of capacity building as a policy intervention is the EEE (Energy and Environmental Expert) scheme, implemented in Finland from 1995 (Wolff *et al.* 2017). This was developed by a state-owned agency but run by volunteers in the community to provide peer-to-peer advice. Training and capacity building on sustainable choices was provided to community volunteers. These volunteers then disseminated the information more widely through acting as a point of expertise to advise other residents in the community on monitoring and reducing environmental issues. Success of the programme was variable across individual cases, but with some cases leading to a significant reduction in water and energy consumption and suggesting longer term effects through changing consumer attitudes (Wolff *et al.* 2017).

In the UK, this could be implemented at any scale (national, devolved, or local government).

4.2.5 Partnership-based interventions

Partnership-based interventions rely on actors working together to exchange knowledge and to progress implementation of sustainable consumption options. Public-private partnerships, where governments work with businesses, are used in many countries. In such cases, governments can get knowledge of how businesses work, whilst businesses feel ownership of inputting to the process. Beginning with this and only implementing regulatory interventions (see Section 4.4) if partnership-based interventions have failed is often seen as a useful stepped approach. Concerns have been raised over the potential for increases in industry influence on food systems, but effective collaborations are designed to obtain gains that would otherwise not be possible (Smyth *et al.* 2021). Partnership-based interventions have been commonly used in attempts to improve public health (Lourenço *et al.* 2019; Pan American Health Organisation 2015), but examples of their use in attempts to improve environmental sustainability also exist.

One such example is the UK the Roundtable on Sustainable Soya (Efeca and Partnership for Forests 2018), which is brought together and funded by the UK Government, and consists of key players in the UK soya market. It convened for the first time in 2017 and aims to support progress towards a goal of soya use that is "legal and cultivated in a way that protects against conversion of forests and valuable native vegetation."

In the UK, this could be implemented by any level of government.

4.3 Economic interventions

Economic interventions aim to make sustainable options more financially viable for supply chain actors. This could be through taxing or reducing subsidies on the most harmful options, reducing taxes or subsidising the most sustainable options, or funding research on and implementation of sustainable solutions. As well as addressing products directly, taxes or subsidies could also be applied to other aspects addressed in the review, such as subsidising the building of sustainable waste infrastructure. Taxes and subsidies could be aimed either at the end consumer to try to change consumption-related behaviours, or at other stages of the supply chain such as to try to change the behaviours of businesses and traders.

Advantages of these kinds of interventions include that they are often seen to be flexible and efficient ways to create change through provision of incentives for behavioural change (Pape *et al.* 2011). The price of a good (alongside quality and income level) is generally one of the key factors influencing purchasing decisions (BIO Intelligence Service 2012). The full cost based on natural capital thinking (including factors such as the cost that would be required to clear up all pollution caused by the good's production) is not currently included in prices paid by consumers in most cases; if it were, there would be a clear signal and motivation to change purchasing decisions (BIO Intelligence Service 2012). Changing purchasing decisions made by consumers may also lead to transformations within businesses who seek to remain competitive by matching this change in demand through more sustainable sourcing.

Disadvantages include that some people have concerns over the level of intervention a government can and should make to what can be seen as consumers' private lives or the interests of individual businesses (Wolff *et al.* 2017). They can also lead to unpredictable and sometimes unequally distributed costs (Pape *et al.* 2011), although this could be mitigated by ensuring prices are progressively graduated based on taxes and supporting

low-income households (BIO Intelligence Service 2012). Whilst research supports the use of financial incentives as a driver of short-term behavioural changes, it has been shown in certain cases to reduce intrinsic motivation for such behavioural changes, which is thought to be an important factor in sustaining changes in the long term (Pape *et al.* 2011).

As there have been fewer examples of economic interventions to encourage sustainable consumption than there are for information-based interventions, there is also less research available on their effectiveness (Wolff *et al.* 2017). The importance of ensuring that economic incentives are distributed throughout the supply chain so all actors share the costs and incentives for change has been highlighted as a key factor for ensuring their success (Jonkutė-Vilkė & Staniškis 2019).

4.3.1 Taxes and fees

Examples of tax or fee based sustainable consumption policies that have been implemented include:

- Ireland's pay-by-use waste charging system, which was found to lead to reductions in waste collection, but also a high level of resistance and tension at its introduction (Pape *et al.* 2011). If implementing similar policies in future, the authors of Pape *et al.* (2011) recommend ensuring a greater level of guidance is provided for waste collectors involved in the policy's implementation and ensuring that the payments required relate to the amount of waste produced to ensure the 'polluter pays' principle is taken into account, rather than just using a flat fee for any collection.
- Vehicle taxes, which have been implemented in a number of countries based on the emissions produced by each type of vehicle (Beacon 2018; UNEP 2012). For example, France introduced the "bonus/malus" system in 2007, in which sales of highly polluting vehicles are taxed, the revenue from which is used to support a credit-based system to subsidise sales of low emission vehicles. It has been considered a very effective policy in terms of shifting sales towards those that produce less pollution, with a rise in the purchase of fuel-efficient vehicles from 30% in 2007 to 56% in 2009 (Beacon 2018; BIO Intelligence Service 2012). Adjustments had to be made to ensure cost neutrality, as there was a bigger response to the incentives than initially predicted (Beacon 2018). Strong engagement with the automotive industry has helped with the policy's success by ensuring understanding and support (Beacon 2018).
- Plastic bag levies or bans have also been introduced in a wide range of countries, including levies in all four countries of the UK. These require consumers to pay a small charge to use plastic bags when shopping, thereby changing the 'default' option. Estimates suggest this has led to a 71–96% reduction in plastic bag use across the four UK countries (Xanthos & Walker 2017).

A number of papers also suggested the possibility of using Life Cycle Assessment (LCA) measurements to reform the Value Added Tax (VAT) system so that it includes ecologically differentiated VAT based on the environmental damages that a good's production is estimated to cause across all sectors of the economy (Bahn-Walkowiak & Wilts 2015; De Camillis & Goralczyk 2013; Timmermans & Achten 2018). Although suggested (and considered potentially feasible) in several scientific papers, no examples of actual implementation of such a policy were found.

In the UK, taxes and fees could be implemented by either national or devolved governments.

4.3.2 Subsidies

Examples of subsidy or payment based sustainable consumption policies that have been implemented include:

- Deposit refund schemes (see Section 4.1.1)
- Incentive schemes for electric cars, which have been adopted in a wide range of countries. In the USA, it has been estimated that for every \$1000 of subsidy offered, sales of electric vehicles increased by between 2.5% and 11% (Jenn *et al.* 2018; Wee *et al.* 2018). In Europe, estimates suggest an increase of 5-7% for every €1000 (Münzel *et al.* 2019). However, other factors such as the provision of charging infrastructure (of relevance to Section 4.1) were found to have a greater effect than incentives alone (Sierzchula *et al.* 2014).
- Reform and reduction of harmful subsidies. For example, in Indonesia, fuel subsidies were reduced (UNEP 2015).

In the UK, subsidies could be implemented by either national or devolved governments.

4.3.3 Funding research and implementation

Innovation focusing on the development of more sustainable production practices and supply chain processes could lead to changes that mean increases in efficiency, with reduced environmental impacts from consumption of the same amount of commodity. Similarly, once research has identified such an improvement, its uptake and implementation will require supply chain actors to provide training and to change their current practices and infrastructure, which is also associated with a cost. Provision of funding to these cases is therefore another policy intervention that could be used by governments to improve the sustainability of their country's consumption (Jonkutė-Vilkė & Staniškis 2019).

This could be executed through existing instruments such as research councils and overseas development assistance, so could be relatively simple to implement as it would only require ensuring that sustainable consumption is specified in calls for projects and assessed against as a key criterion when making funding decisions. Alternatively, it could be implemented on a larger scale through the establishment of a new instrument, such as a research council with a specific sustainability remit.

In the UK, funding could be provided at any scale (national, devolved, or local government).

4.4 Regulatory interventions

Regulatory interventions aim to make use of legislation or commitments made with partner countries in order to reduce the environmental impact of consumption. This may form part of an international or bilateral agreement, a ban or quota, or rules relating to public procurement.

Regulatory interventions are the type that are the most likely to have a direct impact on the behaviour of supply chain actors, as they rely on outright bans rather than nudging behaviour. Whilst other intervention types mentioned often focus more on breaking down barriers for the frontrunners who are already conscious of and making efforts towards sustainable consumption, regulatory interventions may be more effective for actors who are less inclined to strive for positive change.

As with economic interventions, some people have concerns over the level of intervention a government can and should make to what can be seen as consumers' private lives or the interests of individual businesses (Wolff *et al.* 2017); a sentiment that may be stronger for an outright ban than for changes in prices. Bans also run a greater risk of 'black markets' developing, with goods being sold illegally, so implementation of regulatory interventions must be coupled with appropriate enforcement in order to be effective.

Additionally, the implementation of regulatory policies that lead to barriers to trade must ensure they meet WTO (World Trade Organisation) rules. These state that countries cannot ban the imports of goods outright and that tariffs on goods must be the same for all countries for which a specific bilateral trade agreement has not been made. except in a very specific set of cases listed in Article XX of the General Agreement on Tariffs and Trade (GATT). These exceptions include where measures are "necessary to protect public morals," "necessary to protect human, animal or plant life or health" and "relating to the conservation of exhaustible natural resources if such measures are made effective in conjunction with restrictions on domestic production or consumption." This does provide scope for bans on environmental grounds, such as the ban on ozone-depleting products that was made as part of the Montreal Protocol or bans relating to CITES (the Convention on International Trade in Endangered Species of Wild Fauna and Flora). However, it does not provide scope for discrimination between 'like' products, which would enable broader application. A lack of clarity from WTO remains a commonly cited barrier to implementation of these kinds of policy intervention, although a recent WWF report provides a detailed legal analysis of the situation (van der Ven et al. 2022).

Similar to economic interventions, there have been relatively few examples of regulatory interventions to encourage sustainable consumption in comparison with information-based interventions, so there is also less research available on their effectiveness (Wolff *et al.* 2017).

4.4.1 Trade agreements

Free trade agreements take place bilaterally between two countries or trading blocs in order to negotiate trade tariffs. Often, agreements contain text addressing the environmental impacts associated with the trade in question. In many cases, this simply acknowledges that certain activities are environmentally harmful and states that both parties should endeavour to limit impact. In some cases, a more specific analysis of the environmental impacts and sustainability of the trade between two countries is undertaken. Although relatively few mentions of this and no assessments of effectiveness were found within the literature review, encouraging this more detailed analysis nonetheless represents an opportunity more fully to embed environmental negotiations within trade negotiations in a way that meets WTO requirements (i.e. does not need to meet one of the specific exceptions listed above, as it is only applied between the two countries that have agreed on it).

Trade agreements would focus on trade (which covers imports and exports, regardless of whether these are then consumed within the country or processed and re-exported) rather than consumption (final use). However, there is likely to be a strong link between trade and consumption: trade is required to provide sustainable options for consumption even if some of these are re-exported. Additionally, addressing sustainability in trade can help reduce consumption impacts overall at the global scale (regardless of whether the impact is linked to consumption of the country in question or to the consumption of a country that imports the re-exported goods further up the supply chain).

In the UK, this would need to be implemented centrally through national government.

4.4.2 Multilateral environmental agreements (MEAs)

MEAs consist of agreements made between at least three states, which aim to improve environmental outcomes. They include both "soft law" (where parties are obligated to consider defined environmental principles) and "hard law" (where parties must undertake legally binding actions) options. They often include setting targets that are agreed across multiple countries which aim to galvanise action.

The Montreal Protocol is one example of an MEA that has been implemented very effectively. This aimed to ban ozone depleting substances, thereby taking them off the market so that no supply chain actors had access to them, forcing an improvement in the sustainability of their consumption. It was adopted on 15 September 1987 and to this date is the only MEA that has been agreed to by every country of the world. Without the Protocol, it is estimated that ozone depletion would have increased ten times by 2050 and led to an additional two million skin cancer deaths each year by 2030 (UNEP 2018). Because of the treaty, it is estimated that the ozone layer will have recovered by 2050 (UNEP 2018). One factor contributing to the success of the treaty is the Multilateral Fund for the Implementation of the Montreal Protocol, which ensured that all countries were able to take the necessary actions and carry out the required monitoring (UNEP 2012).

Another example of an MEA is the Amsterdam Declarations, made by seven European countries in 2015, which aimed "to lend public sector support to the implementation of existing private and public sector commitments to achieve fully sustainable and deforestation-free agro-commodity supply chains in Europe by 2020" (Amsterdam Declarations Partnership 2016). This target was not met, and sources in the literature are sceptical of the extent of the impact that the agreement has had at all (Ermgassen et al. 2020). Reasons suggested for this include a focus on engagement rather than implementation, a reduction in the market share of the signatory countries as others such as China have grown over the relevant time period, the complexity of the supply chains involved, and the need for a greater understanding of the implications on local stakeholders in producer countries (Ermgassen et al. 2020; Lyons-White et al. 2020). A similar commitment to "working collectively to halt and reverse forest loss and land degradation by 2030 while delivering sustainable development and promoting an inclusive rural transformation" was made in 2021 as the Glasgow Leaders' declaration on forests and land use, this time signed by 145 countries (Messetchkova 2021). Not enough time has passed to understand the impacts of this MEA, but the expanded number of signatories and the more holistic nature of the wording are positives, while the lack of clarity around whether referring to gross or net deforestation has been suggested as a likely limiting factor (Gasser et al. 2022).

One key advantage of MEAs over other policy interventions listed in this document, if adopted by enough partner countries, is the reduction in risk of displacement. A key concern when implementing demand side policy interventions to improve the sustainability of production is whether they will actually make a difference overall, or whether other nations will simply take on the goods rejected or reduced by the country implementing the policies. If a significant number of countries, have signed up to undertake similar actions, the risk of this reduces considerably. MEAs are therefore a key component to include in any suite of policy interventions aimed at sustainable consumption, to ensure the relevance and effectiveness of any others that are deployed.

In the UK, this would need to be implemented centrally through national government.

4.4.3 Banning of or quotas on the least sustainable options

Bans or quotas may focus on removing or reducing a particular option from the market. Bans due to environmental reasons in the field of chemical substances are relatively widespread, such as through the European Union's REACH (Registration, Evaluation, Authorization and restriction of Chemicals) directive (UNEP 2015, 2012). There are far fewer examples of bans relating to other sectors of the economy that affect consumers more directly on a day-to-day basis, such as foodstuffs, although bans relating to the consumption of single use plastics are becoming increasingly popular (UNEP 2015).

Bans or quotas may also take the form of defining specific sustainability standards below which goods are not allowed to enter the market. Again, these are relatively common in certain contexts (such as vehicles in terms of fuel emissions and energy efficiency standards for household appliances), but far less so in others, including food products (UNEP 2015). A number of recent reports and commissions have recommended the use of 'core environmental standards' to define minimum environmental conditions for imports (Department for International Trade 2021; Dimbleby 2021; The Climate Change Committee 2022; WWF 2022).

One example of legislation banning the least sustainable options is the Due Diligence legislation being adopted or considered by the UK, the EU and the USA. These vary slightly in scope (e.g. whether all deforestation is considered, or only deforestation that is illegal in the producer country) but each aim to define unacceptable deforestation and obligate companies above a certain size to ensure that they have performed Due Diligence on their supply chains to guarantee that what they are importing does not contain goods from land that has undergone such deforestation. None of these policies have yet been in place long enough (or are not yet in place) for an assessment of their effectiveness.

An example of a restriction that was commonly mentioned in the scientific literature as something that would lead to significant improvements in the sustainability of consumption, but which has not yet been implemented in any real-life examples, is the concept of restricting meat consumption (González *et al.* 2011; Röös *et al.* 2021; Tucker 2018). This topic is particularly sensitive and controversial as many people see meat consumption as part of their identity (Röös *et al.* 2021). Approaches that encourage 'less and better' meat consumption are therefore likely to be supported and implemented by a wider range of actors than an outright ban (Trewern *et al.* 2022). Similarly, restrictions on packaging sizes were suggested in the literature as a potential mechanism to prevent overconsumption, which would have both environmental and health benefits (Röös *et al.* 2021).

Disadvantages of regulatory approaches include that they may need significant research before implementation (it is important to consider potential economic impacts, alternative options that will be available to consumers, potential for unintended consequences and public acceptance), and they may lead to a need for long term enforcement and administration costs (BIO Intelligence Service 2012; Pape *et al.* 2011; UNEP 2015).

In the UK, bans or quotas could be implemented by either national or devolved governments.

4.4.4 Sustainable public procurement rules

Another way governments can influence the sustainability of consumption is through what they consume themselves. As public procurement of goods and services is estimated to make up 13–20% of GDP in developed countries (Fuentes Bargues *et al.* 2021), governments have significant power to implement change through rules and requirements

surrounding their own purchasing practices (Jonkutė-Vilkė & Staniškis, 2019). This could be considered a regulatory intervention because it involves defining rules, but it would also have an economic influence through supporting sustainable businesses and an information-based influence in terms of setting a good example for others to follow. For example, if school meals are sustainable and this is combined with interventions discussed in section 4.2, they may get more ideas about what to cook at home as an adult. The set of rules developed could also be extended voluntarily into the private sector, so that sustainably minded shops or restaurants could use it as a model to follow, if the rules and requirements are made public and easy to understand.

One example of a sustainable public procurement scheme is the Netherlands' CO_2 Performance Ladder. This uses certification (see Section 4.2.1) to give companies a competitive advantage over those that have not been certified. One study estimated that it has the potential to reduce total Dutch CO_2 emissions by 0.8–1.5% per year (Rietbergen & Blok 2013). Another study estimated that total CO_2 reductions between 2010 and 2013 in the Netherlands that could be attributed to the Performance Ladder were 1–1.6% (Rietbergen *et al.* 2017).

However, another study, which looked at Swedish cleaning service procurements (a case where very detailed environmental standards are set) found limited effect of green public procurement on supplier behaviours (Lundberg *et al.* 2015). It was suggested that the total number of bidders and their decision to participate in procurement processes are key factors to consider when determining how effective sustainable public procurement rules will be (Lundberg *et al.* 2015). Another paper has pointed out trade-offs that must be considered, relating to factors such as the higher price of sustainable options conflicting with parallel aims about keeping public sector costs as low as possible, and additional hygiene regulations that would need to be followed if switching from high impact pre-processed foods to raw foodstuffs (Wolff *et al.* 2017).

In the UK, this could be implemented at any scale (national, devolved, or local government).

4.4.5 Controls on advertising

One suggestion in the literature was that governments could provide regulations on advertising of the most harmful products (UNEP 2015). No examples of actual implementation in the context of sustainable consumption were found, but a similar intervention has been widely used for products that can be damaging to health, such as for cigarettes and alcohol. Marketing and media are known to have significant impacts on consumer behaviour and choices, so working with those in this sector and ensuring that advertisements are not over-aggressive, promoting overconsumption, or focused on the least sustainable options could help to encourage people to move toward more sustainable choices, especially if combined with public information campaigns (UNEP, 2012).

In the UK, this could be implemented by either national or devolved governments.

5 Conclusions

Overall, there are many different policy interventions (Section 4) that can be targeted at different actors (Section 2) and can focus on breaking down different barriers to sustainable consumption (Section 3). Each intervention, actor or barrier alone will do little to contribute to the overall problem if carried out in isolation; an effective solution will require a holistic integration of as many of these factors as possible, and a ratcheting of policies that start small and then build in ambition to address the systemic nature of the problem. Further research is required in most cases to demonstrate how effective each potential policy intervention is, what synergies they bring to each other and what contexts they work best within. A greater understanding of the potential for perverse incentives, such as displacement of impacts to other actors, will also be necessary, although the relevance of this will reduce as the coverage of such interventions increases globally. It will also be important to consider sustainable consumption policy interventions within the wider context of other factors, such as combining environmental factors with health factors when encouraging a change in consumption patterns. Learning from other areas, such as health, which are more advanced in their thinking on using policy to create cultural change (e.g. The Nuffield Council on Bioethics 2007), would also be useful.

References

Amsterdam Declarations Partnership. 2016. Strategy AD Partnership.

Bahn-Walkowiak, B. & Wilts, H. 2015. Reforming the EU VAT system to support the transition to a low-carbon and resource efficient economy, Carbon Pricing. Edward Elgar Publishing.

Beacon. 2018. Factsheet: Bonus-Malus Vehicle Incentive System (France). EUKI. URL <u>https://www.euki.de/en/euki-publications/factsheet-bonus-malus-vehicle-incentive-system-france/</u> (accessed 8.23.22).

BIO Intelligence Service. 2012. Policies to encourage sustainable consumption, Final report prepared for European Commission (DG ENV).

Burgess, M., Holmes, H., Sharmina, M. & Shaver, M.P. 2021. The future of UK plastics recycling: One Bin to Rule Them All. Resources, *Conservation and Recycling* 164, 105191. <u>https://doi.org/10.1016/j.resconrec.2020.105191</u>

Croft, S., West, C., Harris, M., Green, J., Molotoks, A., Harris, V. & Way, L. 2021. Technical documentation for an experimental statistic estimating the global environmental impacts of UK consumption. (No. JNCC Report No. 695). JNCC, Peterborough, ISSN 0963-8091. https://webarchive.nationalarchives.gov.uk/ukgwa/20220901105554/https://hub.jncc.gov.uk/assets/91efc19d-f675-426f-9333-ed0195cc729d

De Camillis, C. & Goralczyk, M. 2013. Towards stronger measures for sustainable consumption and production policies: proposal of a new fiscal framework based on a life cycle approach. Int J Life Cycle Assess 18, 263–272. <u>https://doi.org/10.1007/s11367-012-0460-5</u>

Deloitte. 2022. Sustainability & Consumer Behaviour 2022 [WWW Document]. Deloitte United Kingdom. <u>https://www2.deloitte.com/uk/en/pages/consumer-business/articles/sustainable-consumer.html</u> (accessed 9.1.22).

Department for International Trade. 2021. Trade and Agriculture Commission: Final Report (executive summary) [WWW Document]. <u>https://www.gov.uk/government/publications/trade-and-agriculture-commission-tac/trade-and-agriculture-commission-final-report-executive-summary</u> (accessed 2.20.23).

Deselnicu, D., Vasilescu, A., Purcarea, A. & Militaru, G. 2014. Sustainable Consumption and Production in the Footwear Sector - ProQuest. *Revista de Pielarie Incaltaminte* 14, 159–180.

Dimbleby. 2021. The National Food Strategy - The Plan [WWW Document]. National Food Strategy. <u>https://www.nationalfoodstrategy.org/</u> (accessed 2.20.23).

Efeca & Partnership for Forests. 2018. UK Roundtable on Sustainable Soya: Baseline study 2018.

FAO. 2011. Global food losses and food waste - Extent, causes and prevention. Food and Agriculture Organisation of the United Nations, Rome.

Fuentes Bargues, J., Piccirillo, E. & Rebaudengo, M. 2021. Green Public Procurement of Building Works in the Piedmont Region of Italy.

Gasser, T., Ciais, P. & Lewis, S.L. 2022. How the Glasgow Declaration on Forests can help keep alive the 1.5°C target. *Proceedings of the National Academy of Sciences* 119, e2200519119. <u>https://doi.org/10.1073/pnas.2200519119</u>

Geyer, R., Jambeck, J. & Law, K. 2017. Production, use, and fate of all plastics ever made. *Science Advances* 3, e1700782. <u>https://doi.org/10.1126/sciadv.1700782</u>

González, A.D., Frostell, B. & Carlsson-Kanyama, A. 2011. Protein efficiency per unit energy and per unit greenhouse gas emissions: Potential contribution of diet choices to climate change mitigation. *Food Policy* 36, 562–570. <u>https://doi.org/10.1016/j.foodpol.2011.07.003</u>

Gosar, M. 2004. Environmental impacts of metal mining= Vplivi kovinskih rudnikov na okolje 51, 2097–2107.

Graça, J., Godinho, C.A. & Truninger, M. 2019. Reducing meat consumption and following plant-based diets: Current evidence and future directions to inform integrated transitions. *Trends in Food Science & Technology* 91, 380–390. https://doi.org/10.1016/j.tifs.2019.07.046

Iraldo, F. & Barberio, M. 2017. Drivers, Barriers and Benefits of the EU Ecolabel in European Companies' Perception. *Sustainability* 9, 751. <u>https://doi.org/10.3390/su9050751</u>

Jenn, A., Springel, K. & Gopal, A.R. 2018. Effectiveness of electric vehicle incentives in the United States. *Energy Policy* 119, 349–356. <u>https://doi.org/10.1016/j.enpol.2018.04.065</u>

Jonkutė-Vilkė, G. & Staniškis, J. 2019. The Role of Different Stakeholders in Implementing Sustainable Consumption and Production in Lithuania. *Environmental engineering and management journal* 18, 617–632. <u>https://doi.org/10.30638/eemj.2019.057</u>

Lorek, S. & Fuchs, D. 2013. Strong sustainable consumption governance – precondition for a degrowth path? Journal of Cleaner Production, Degrowth: *From Theory to Practice* 38, 36–43. <u>https://doi.org/10.1016/i.jclepro.2011.08.008</u>

Lourenço, S., Laub Hansen, G., Stærk, B., Frank, P. & Toft Petersen, C. 2019. The Whole Grain Partnership - How a Public-Private Partnership Helped Increase Whole Grain Intake in Denmark.

Lundberg, S., Marklund, P., Strömbäck, E. & Sundström, D. 2015. Using public procurement to implement environmental policy: an empirical analysis | SpringerLink [WWW Document]. <u>https://link.springer.com/article/10.1007/s10018-015-0102-9</u> (accessed 8.26.22).

Lyons-White, J., Pollard, E.H.B., Catalano, A.S. & Knight, A.T. 2020. Rethinking zero deforestation beyond 2020 to more equitably and effectively conserve tropical forests. *One Earth* 3, 714–726. <u>https://doi.org/10.1016/j.oneear.2020.11.007</u>

Martini, A. 2021. Socially responsible investing: from the ethical origins to the sustainable development framework of the European Union. *Environ Dev Sustain* 23, 16874–16890. https://doi.org/10.1007/s10668-021-01375-3

Messetchkova, I. 2021. Glasgow Leaders' Declaration on Forests and Land Use [WWW Document]. UN Climate Change Conference (COP26) at the SEC – Glasgow 2021. <u>https://ukcop26.org/glasgow-leaders-declaration-on-forests-and-land-use/</u> (accessed 8.26.22). Mirela, P., Caraman, I., Nedeff, V., Inglezakis, V., Venetis, C., Coutsikos, P., Lazar, G. & Bârsan, N. 2014. Green market: A comparative study in Romania. *Fresenius Environmental Bulletin* 23.

Münzel, C., Plötz, P., Sprei, F. & Gnann, T. 2019. How large is the effect of financial incentives on electric vehicle sales? – A global review and European analysis. *Energy Economics* 84, 104493. <u>https://doi.org/10.1016/j.eneco.2019.104493</u>

Neill, P. 2020. Majority of business owners want to be more environmentally sustainable. *Environment Journal*. <u>https://environmentjournal.online/articles/majority-of-business-owners-want-to-be-more-environmentally-sustainable/</u> (accessed 9.1.22).

Pan American Health Organisation. 2015. Public-Private Partnerships with the Food Industry.

Pape, J., Rau, H., Fahy, F. & Davies, A. 2011. Developing Policies and Instruments for Sustainable Household Consumption: Irish Experiences and Futures. *Journal of Consumer Policy* 34, 25–42. <u>https://doi.org/10.1007/s10603-010-9151-4</u>

Pendrill, F., Persson, U.M., Godar, J., Kastner, T., Moran, D., Schmidt, S. & Wood, R. 2019. Agricultural and forestry trade drives large share of tropical deforestation emissions. *Global Environmental Change* 56, 1–10. <u>https://doi.org/10.1016/j.gloenvcha.2019.03.002</u>

Poore, J. & Nemecek, T. 2018. Reducing food's environmental impacts through producers and consumers. *Science* 360, 987–992. <u>https://doi.org/10.1126/science.aaq0216</u>

Prieto-Sandoval, V., Mejía-Villa, A., Ormazabal, M. & Jaca, C. 2020. Challenges for ecolabeling growth: lessons from the EU Ecolabel in Spain. *Int J Life Cycle Assess* 25, 856–867. <u>https://doi.org/10.1007/s11367-019-01611-z</u>

Raja, A.S.M., Arputharaj, A., Saxena, S. & Patil, P.G. 2019. 9 - Water requirement and sustainability of textile processing industries, in: Muthu, S.S. (Ed.), Water in Textiles and Fashion. Woodhead Publishing, pp. 155–173. <u>https://doi.org/10.1016/B978-0-08-102633-5.00009-9</u>

Rietbergen, M.G. & Blok, K. 2013. Assessing the potential impact of the CO2 Performance Ladder on the reduction of carbon dioxide emissions in the Netherlands. *Journal of Cleaner Production* 52, 33–45. <u>https://doi.org/10.1016/j.jclepro.2013.03.027</u>

Rietbergen, M.G., Opstelten, I.J. & Blok, K. 2017. Improving energy and carbon management in construction and civil engineering companies—evaluating the impacts of the CO2 Performance Ladder. *Energy Efficiency* 10, 55–79. <u>https://doi.org/10.1007/s12053-016-9436-9</u>

Röös, E., Larsson, J., Sahlin, K., Jonell, M., Lindahl, T., André, E., Säll, S., Harring, N. & Persson, M. 2021. Policy Options for Sustainable Food Consumption – Review and Recommendations for Sweden. Mistra Sustainable Consumption report 1:10. Chalmers university of Technology.

Scholl, G., Rubik, F., Kalimo, H., Biedenkopf, K. & Söebech, Ó. 2010. Policies to Promote Sustainable Consumption: Innovative Approaches in Europe. *Natural Resources Forum* 34, 39–50. <u>https://doi.org/10.1111/j.1477-8947.2010.01294.x</u>

Sierzchula, W., Bakker, S., Maat, K. & van Wee, B. 2014. The influence of financial incentives and other socio-economic factors on electric vehicle adoption. *Energy Policy* 68, 183–194. <u>https://doi.org/10.1016/j.enpol.2014.01.043</u>

Smyth, S.J., Webb, S.R. & Phillips, P.W.B. 2021. The role of public-private partnerships in improving global food security. *Global Food Security* 31, 100588. <u>https://doi.org/10.1016/j.gfs.2021.100588</u>

Stevens, C. 2010. Linking sustainable consumption and production: The government role. *Natural Resources Forum* 34, 16–23. <u>https://doi.org/10.1111/j.1477-8947.2010.01273.x</u>

Struk, M. 2017. Distance and incentives matter: The separation of recyclable municipal waste. *Resources, Conservation and Recycling* 122, 155–162. <u>https://doi.org/10.1016/j.resconrec.2017.01.023</u>

Testa, F., Iraldo, F., Vaccari, A. & Ferrari, E. 2015. Why Eco-labels can be Effective Marketing Tools: Evidence from a Study on Italian Consumers. *Business Strategy and the Environment* 24, 252–265. <u>https://doi.org/10.1002/bse.1821</u>

The Climate Change Committee. 2022. Trade policies and emissions reduction: Establishing and assessing options in agriculture and deforestation (Ricardo Energy & Environment). Climate Change Committee. <u>https://www.theccc.org.uk/publication/trade-policies-and-emissions-reduction-establishing-and-assessing-options-in-agriculture-and-deforestation-ricardo-energy-environment/</u> (accessed 2.20.23).

The Nuffield Council on Bioethics. 2007. Policy process and practice [WWW Document]. The Nuffield Council on Bioethics. <u>https://www.nuffieldbioethics.org/publications/public-health/guide-to-the-report/policy-process-and-practice</u> (accessed 2.20.23).

Timmermans, B. & Achten, W.M.J. 2018. From value-added tax to a damage and valueadded tax partially based on life cycle assessment: principles and feasibility. *Int J Life Cycle Assess* 23, 2217–2247. <u>https://doi.org/10.1007/s11367-018-1439-7</u>

Trewern, J., Chenoweth, J. & Christie, I. 2022. "Does it change the nature of food and capitalism?" Exploring expert perspectives on public policies for a transition to 'less and better' meat and dairy. *Environmental Science & Policy* 128, 110–120. <u>https://doi.org/10.1016/j.envsci.2021.11.018</u>

Tucker, C. 2018. Using environmental imperatives to reduce meat consumption: perspectives from New Zealand. Kōtuitui: *New Zealand Journal of Social Sciences Online* 13, 99–110. <u>https://doi.org/10.1080/1177083X.2018.1452763</u>

UNEP. 2021. Food Waste Index Report. United Nations Environment Programme, Nairobi.

UNEP. 2018. The Montreal Protocol [WWW Document]. Ozonaction. http://www.unep.org/ozonaction/who-we-are/about-montreal-protocol (accessed 8.26.22).

UNEP. 2017. Sustainable consumption and production policies [WWW Document]. UNEP - UN Environment Programme. <u>http://www.unep.org/explore-topics/resource-efficiency/what-we-do/sustainable-consumption-and-production-policies</u> (accessed 8.9.22).

UNEP. 2015. Sustainable Consumption Guide for Policymakers: Debunking Myths and Outlining Solutions (Asia Edition). United Nations Environment Programme, Bangkok.

UNEP. 2012. Global Outlook on Sustainable Consumption and Production Policies: Taking action together: Sustainable Development Knowledge Platform [WWW Document]. <u>https://sustainabledevelopment.un.org/index.php?page=view&type=400&nr=559&menu=151</u> <u>5</u> (accessed 8.11.22).

Upham, P., Dendler, L. & Bleda, M. 2011. Carbon labelling of grocery products: public perceptions and potential emissions reductions. *Journal of Cleaner Production* 19, 348–355. <u>https://doi.org/10.1016/j.jclepro.2010.05.014</u>

Urban Sustainability Exchange. 2014. "The Sharing City, Seoul" Project [WWW Document]. <u>https://use.metropolis.org/case-studies/the-sharing-city-seoul-project</u> (accessed 8.11.22).

van der Ven, C., Antoni, E., Moraa Mokaya, M., Buckwell, A. & Nicholson, M. 2022. Core Environmental Standards for UK imported Agri Food Products: Options for Pesticide and Fertiliser Use. WWF and Tulip Consulting.

Webb, J., Williams, A.G., Hope, E., Evans, D. & Moorhouse, E. 2013. Do foods imported into the UK have a greater environmental impact than the same foods produced within the UK? Int J Life Cycle Assess 18, 1325–1343. <u>https://doi.org/10.1007/s11367-013-0576-2</u>

Wee, S., Coffman, M. & La Croix, S. 2018. Do electric vehicle incentives matter? Evidence from the 50 U.S. states. Research Policy 47, 1601–1610. https://doi.org/10.1016/j.respol.2018.05.003

Wolff, F., Schönherr, N. & Heyen, D.A. 2017. Effects and success factors of sustainable consumption policy instruments: a comparative assessment across Europe. Journal of Environmental Policy & Planning 19, 457–472. https://doi.org/10.1080/1523908X.2016.1254035

WRAP. 2022. Redesigning the plastics system. Waste and Resources Action Programme, Banbury.

WWF. 2022. Setting the standard [WWW Document]. WWF. <u>https://www.wwf.org.uk/our-reports/setting-standard-report</u> (accessed 2.20.23).

WWF UK. 2021. Driven to Waste: The Global Impact of Food Loss and Waste on Farms. Woking.

Xanthos, D. & Walker, T.R. 2017. International policies to reduce plastic marine pollution from single-use plastics (plastic bags and microbeads): A review. Marine Pollution Bulletin 118, 17–26. <u>https://doi.org/10.1016/j.marpolbul.2017.02.048</u>

zu Ermgassen, E.K.H.J., Ayre, B., Godar, J., Lima, M.G.B., Bauch, S., Garrett, R., Green, J., Lathuillière, M.J., Löfgren, P., MacFarquhar, C., Meyfroidt, P., Suavet, C., West, C. & Gardner, T. 2020. Using supply chain data to monitor zero deforestation commitments: an assessment of progress in the Brazilian soy sector. Environ. Res. Lett. 15, 035003. https://doi.org/10.1088/1748-9326/ab6497

Annex 1: Methods

This report was undertaken as a rapid, time-limited (five days total for reading and write-up) synthesis of a literature review of scientific papers and the grey literature. The review did not aim to be comprehensive, but rather to provide a useful amount of information to frame key opportunities and issues.

A Scopus search was made using the syntax: "policy lever*" OR "policy action*" AND "sustainable consumption" OR "sustainable trade". This returned 386 results, 104 of which were considered of relevance based on abstract screening. Of these, the first 24 on the list were read and summarised in note form to input into the synthesis. Others were excluded due to time constraints.

A search for "[country name] sustainable consumption policy" was subsequently made for Australia, Canada, the USA, New Zealand, China, Germany, France, Spain, Sweden, and Finland. These countries were selected subjectively, based on factors such as the author's prior knowledge of those that have implemented sustainable consumption policies, those with large consumption impacts and those that may sit within similar context to the UK. This process led to the identification of 55 relevant papers and reports, of which information from a subset of eight documents identified was recorded, again largely restricted by time constraints.

The information from these papers was then synthesised into this report.

This study did not directly engage with any of the bodies within the UK who would be in position to implement such policies (although some kindly reviewed the report), which would be an obvious next step in the research.